Mauna Loa Solar Observatory Observer's Log ______ Wed Nov 1 16:37:55 GMT 2017 Year: 17 Doy: 305 Observer: berkey WEATHER COMMENT: berkey: Wed Nov 01 16:39:14 GMT 2017 Temp: 39.1f, Humidity: 58%, Pressure: 28.789in, Wind: 8mph from 185degs, Skies: clear with high inverison layer likely to see a bit of haze in the coming hours. end Wed Nov 01 18:55:47 GMT 2017 CoMP occulter has been re-centered Wed Nov 01 18:58:37 GMT 2017 SGS Alignment complete GENERAL COMMENT BY berkey: Wed Nov 01 18:57:10 GMT 2017 PM Blew off Kcor O1 end GENERAL COMMENT BY berkey: Wed Nov 01 18:57:14 GMT 2017 PM Blew off CoMP 01 end Wed Nov 01 19:08:44 GMT 2017 CoMP occulter has been re-centered Wed Nov 01 19:08:45 GMT 2017 COMP Start Patrol on cookbook: synoptic-00010.cbk Wed Nov 01 19:16:06 GMT 2017 Kcor Focus/alignment program exited Wed Nov 01 19:16:20 GMT 2017 Kcor Focus/alignment program exited Wed Nov 01 19:57:49 GMT 2017 KCOR Start Synoptic Patrol Wed Nov 01 20:30:41 GMT 2017 KCOR End Patrol Wed Nov 01 20:31:56 GMT 2017 CoMP Paused for clouds Wed Nov 01 20:34:05 GMT 2017 CoMP Restarted from pause Wed Nov 01 20:36:33 GMT 2017 CoMP Paused for clouds Wed Nov 01 20:44:10 GMT 2017 CoMP Restarted from pause Wed Nov 01 20:57:33 GMT 2017 CoMP Paused for clouds Wed Nov 01 21:01:31 GMT 2017 CoMP Restarted from pause Wed Nov 01 21:06:16 GMT 2017 CoMP Paused for clouds GENERAL COMMENT BY berkey: Wed Nov 01 22:19:02 GMT 2017 Dome closed for low clounds. end GENERAL COMMENT BY berkey: Thu Nov 02 01:05:02 GMT 2017 New UPS arrived this morning. Boxes were removed from the pallete and brought in the building. UPS's were not unboxed wa iting on 2 staff memebers to be on site before this happens. Still tying to figure out the ultimate location for the UPS we talked about rack mounting; but I dont see any rails in the boxes. A good (maybe better) alternatiive will be to put it on a dolly next to or under the Martix 5000 UPS. end

Thu Nov 02 01:08:23 GMT 2017 CoMP Restarted from pause

Thu Nov 02 01:08:23 GMT 2017 COMP End Patrol